

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Sugiharti, Lilik; Padilla, Miguel Angel Esquivias; Mohd Shahidan Shaari; Jayanti, Ari Dwi; Bin Ridzuan, Abdul Rahim

# Article

# Indonesia's poverty puzzle: Chronic vs. transient poverty dynamics

**Cogent Economics & Finance** 

# **Provided in Cooperation with:**

**Taylor & Francis Group** 

*Suggested Citation:* Sugiharti, Lilik; Padilla, Miguel Angel Esquivias; Mohd Shahidan Shaari; Jayanti, Ari Dwi; Bin Ridzuan, Abdul Rahim (2023) : Indonesia's poverty puzzle: Chronic vs. transient poverty dynamics, Cogent Economics & Finance, ISSN 2332-2039, Taylor & Francis, Abingdon, Vol. 11, Iss. 2, pp. 1-20,

https://doi.org/10.1080/23322039.2023.2267927

This Version is available at: https://hdl.handle.net/10419/304238

## Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

## Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.









**Cogent Economics & Finance** 

ISSN: (Print) (Online) Journal homepage: www.tandfonline.com/journals/oaef20

# Indonesia's poverty puzzle: Chronic vs. transient poverty dynamics

Lilik Sugiharti, Miguel Angel Esquivias, Mohd Shahidan Shaari, Ari Dwi Jayanti & Abdul Rahim Ridzuan

**To cite this article:** Lilik Sugiharti, Miguel Angel Esquivias, Mohd Shahidan Shaari, Ari Dwi Jayanti & Abdul Rahim Ridzuan (2023) Indonesia's poverty puzzle: Chronic vs. transient poverty dynamics, Cogent Economics & Finance, 11:2, 2267927, DOI: <u>10.1080/23322039.2023.2267927</u>

To link to this article: <u>https://doi.org/10.1080/23322039.2023.2267927</u>

© 2023 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group.



0

Published online: 12 Oct 2023.

C	ß
_	

Submit your article to this journal  $\square$ 

Article views: 1471



View related articles 🗹

	Vi
rossMark	

View Crossmark data 🗹



Received: 17 July 2023 Accepted: 03 October 2023

\*Corresponding author: Miguel Angel Esquivias, Faculty of Economics and Business, Universitas Airlangga, Surabaya, Indonesia E-mail: miguel@feb.unair.ac.id

Reviewing editor: Goodness Aye, Agricultural Economics, University Of Agriculture, Makurdi Benue State, Nigeria

Additional information is available at the end of the article

# **DEVELOPMENT ECONOMICS | RESEARCH ARTICLE**

# Indonesia's poverty puzzle: Chronic vs. transient poverty dynamics

💥 cogent

economics & finance

Lilik Sugiharti<sup>1</sup>, Miguel Angel Esquivias<sup>1</sup>\*, Mohd Shahidan Shaari<sup>2</sup>, Ari Dwi Jayanti<sup>1</sup> and Abdul Rahim Ridzuan<sup>3,4,5,6,7</sup>

Abstract: Indonesia has lowered the total poverty rate by less than 10%. Earlier poverty measurements in Indonesia suggest that transient poverty is more prevalent. We argue that, when employing the Equally Distributed Equivalent (EDE) approach and disaggregated poverty lines, chronic poverty is more prevalent than transient poverty. We estimated chronic and transient poverty in Indonesia from 2007 to 2014 by employing a large longitudinal dataset and disaggregated poverty line measures at the district level. The empirical results are robust in various groups based on education, gender, marital status, location (urban-rural), and employment characteristics (status, farming and non-farming, type, and sector). The results indicate that chronic poverty accounts for at least two-thirds of total poverty. Poverty gaps based on education, regional location, gender, and employment are significant. Moreover, we assess whether poverty is linked to socioeconomic aspects and policy programs using quantile regression. The findings indicate that gender (female), age, number of household members, and household location are positively related to higher poverty and chronic poverty. Household head deaths and physical disabilities are positively associated with poverty. Although the urban-rural poverty gap has decreased, casual workers remain prone to poverty. Moreover, poverty is negatively linked to educational attainment, access to financial, transportation, and communication services, suggesting that improving these aspects may help reduce poverty. Social aid programs that support health, food assistance, education, and conditional cash transfers are negatively linked to both total and chronic poverty. Energy subsidies were not associated with lower levels of poverty.

Subjects: Development Policy; Rural Development; Economics and Development; Population & Development; Economics

Keywords: Chronic poverty; transient poverty; gender gap; social welfare programs; poverty alleviation; social assistance; health system access; educational access

JEL Classification: I32; O53

# 1. Introduction

This study provides new evidence of poverty dynamics using the EDE approach applied to Indonesia from 2007 to 2014. Poverty alleviation requires collaborative efforts, including estimation and analysis of deprivation. Within the poverty dimension, decomposition into chronic and





© 2023 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. The terms on which this article has been published allow the posting of the Accepted Manuscript in a repository by the author(s) or with their consent. transient components has emerged as a tool for supporting policymaking. Chronic poverty refers to a long-term state of deprivation (Chung & Maguire-Jack, 2020), while transient poverty is temporary and characterized by individuals entering or leaving it (Dang & Dabalen, 2019). Chronic and transient poverty demand distinct policy responses. The former, which is more closely linked to human and capital development, necessitates long-term plans and structural changes (Sugiharti et al., 2023). The latter, which is of a temporary nature, can be addressed through short-term policies (Tsiboe et al., 2023). Hence, to reduce poverty effectively, it is crucial to decompose poverty into its two components, as they are characterized by distinct underlying causes, durations, and levels of vulnerability (Aikaeli et al., 2021), necessitating tailored approaches to address their root issues and providing effective support (Leal Filho et al., 2022).

This study provides new evidence on poverty decomposition measurements in Indonesia using the EDE approach. Earlier measurements of poverty in Indonesia and other countries include headcount ratios, components, and spell approaches, and tend to report substantially larger rates of transient poverty than chronic ones (Table 1). We argue that the method and measurement of poverty lines matter to appropriately break poverty down into its components. This study applies the EDE approach (Duclos et al., 2010) to decompose poverty for several reasons. First, the results can be compared with commonly used approaches in Indonesia and other developing countries, mainly based on the work of Jalan and Ravallion (1998). Second, the EDE approach allows the breakdown of poverty (chronic and transient components) into the "average poverty gap and cost of poverty inequality" (Mai & Mahadevan, 2016) among individuals to identify the main sources of poverty (Hauser, 2023), which refers to the average poverty gap as the distance between a poverty line and the average income of poor individuals. Meanwhile, the cost of poverty inequality is the decrease in community welfare due to poverty and inequality. Both concepts are critical to the design of welfare redistribution and poverty eradication programs (Ichwara et al., 2023; Poy, 2023). Third, the EDE approach can correct the possible bias arising from limited waves in the data. Such considerations are important in longitudinal studies that use limited data for the analysis.

Table 1 summarizes studies on poverty dynamics in Indonesia and other countries. Most previous studies in Indonesia have employed the spell method (Dartanto & Nurkholis, 2013; Dartanto et al., 2020; Moeis et al., 2020), squared poverty gaps (Bella & Dartanto, 2018; Taufiq & Dartanto, 2020), or headcount ratios (Leeuwen & Földvári, 2016; Mahadevan et al., 2017). Few studies have employed the components approach in Indonesia, except for those by Akita and Dariwardani (2013), who used the Socioeconomic Survey (SUSENAS) dataset, and Dartanto et al. (2020), who used the Indonesian Family Life Survey (IFLS), the same dataset in this study, but the poverty lines were measured at the province level. However, poverty measurements in Indonesia consistently result in a larger transient than chronic poverty component (Mai & Mahadevan, 2016). We provide new estimates of poverty using the EDE approach (Duclos et al., 2010) by employing poverty line measurements at the regency level and in sub-groups of individuals.

Three challenges have been identified in the literature on poverty in Indonesia and other developing countries. First, a substantial number of studies claim that chronic poverty, such as South Africa (Beegle et al., 2016; Schotte et al., 2022) is less prevalent than transient poverty (Table 1). A shortcoming of earlier poverty measurements has been pointed out by Mai and Mahadevan (2016) who noted that earlier poverty measurements underestimated chronic poverty. Second, most studies in Indonesia and other developing nations use national or provincial poverty lines (Sugiharti et al. 2023). We suggest using more detailed poverty lines to estimate poverty elements and chronic poverty sources (the average poverty gap and inequality cost). Income disparity persists across and within provinces in Indonesia (Aginta et al., 2021; Firdausy & Budisetyowati, 2022; Hanandita & Tampubolon, 2016; Vidyattama, 2013). Adopting a poverty line (e.g., province-level or urban-rural levels) below the cost of living in regencies or districts will have implications for estimations of poverty, the poverty gap, and the cost of inequality. Other developing countries have also recorded intra-provincial welfare differences (Lagakos, 2020; Zhang

Table 1. Select	ed studies in po	verty dynamics	(proportion of c	hronic and tran	sient poverty)
Reference	Country	Period	Approach	Chronic	Transient
Ahmed and Tauseef (2022)	Bangladesh	2011-2019	Spells	19%	81%
Farooq and Ahmad (2020)	Pakistan	2001, 2004, 2010	Spells growth incidence curves (GIC)	43%	57%
Shabnam et al. (2023)	Pakistan	2010-2016	Headcount*	45%	55%
Arif et al. (2011)	Pakistan	1998–2001 2001–2004	Headcount ratio	24%	75%
Bayudan- Dacuycuy and Lim (2014)	Philippines	2004-2008	Spells	27%	73%
Villa and Niño-Zarazúa (2019)	Mexico	2002-2011	Spells***	26%	74%
Fernández- Ramos et al. (2016)	Mexico	2002-2005	Spells	36%	64%
Garza- Rodriguez et al. (2010)	Mexico	2002-2005	Components	69%	31%
Aikaeli et al. (2021)	Tanzania	2012-2018	Spells*	30%	70%
Dang and Dabalen (2019)	Sub-Saharan Africa**		Headcount*	33%	67%
Schotte et al. (2022)	South Africa	2008 -2017	Spells	50%	50%
Krishna and Shariff (2011)	India	1993–94 to 2004–05	Headcount	30%	70%
Abanokova et al. (2022)	Russia	1994-2017	Equivalence scales	45%	55%
Radeny et al. (2012)	Kenya	2000–2009	headcount, poverty gap, and poverty severity	22%	78%
Muyanga et al. (2013)	Kenya	1997-2007		63%	37%
Purwono et al. (2021)	Indonesia	2008 - 2010	EDE	92%	8%
Lo Bue and Palmisano (2020)	Indonesia	1993 - 2014	Spells*	20%	80%
			Spells	29%	71%
Dartanto et al. (2020)	Indonesia	1993-2014	Spell	12%	88%
Mai and Mahadevan (2016)	Indonesia	1993-2007	EDE	76%	24%
Akita and Dariwardani (2013)	Indonesia	2008–2010	Component	35%	65%
Dartanto and Nurkholis (2013)	Indonesia	2005 - 2007	Spells	19%	81%

\*Synthetic panel methods, \*\* Regional average of 21 countries covering two periods between 1998 and 2011 (not all countries have same datasets), \*\*\* Markovian approach

et al., 2019). Moreover, Lee et al. (2017) and Glewwe (2012) noted that nearly 15% to 42% of transient poverty reported in earlier studies (i.e., cases in Vietnam and South Korea) results from inconsistent measurements of income or consumption.

Third, earlier studies measuring poverty dynamics for groups of individuals, that is, based on education, provincial location (Akita & Dariwardani, 2013; Purwono et al., 2021), urban-rural (Dartanto & Nurkholis, 2013), job status (Taufiq & Dartanto, 2020), religious background (Mai & Mahadevan, 2016), and age groups (Landiyanto, 2021), mostly adopt poverty lines at the provincial, urban and rural, or national level. This empirical evidence can be extended by using more disaggregated data (i.e., the poverty line at the regency level), which also estimates the poverty gap and the cost of inequality. Previous studies have used poverty lines only at the urban or regional level to examine issues in Africa (Dang & Dabalen, 2019), Kenya (Muyanga et al., 2013), Tanzania (Aikaeli et al., 2021), Pakistan (Farooq & Ahmad, 2020), the Philippines (Bayudan-Dacuycuy & Lim, 2014), Mexico (Fernández-Ramos et al., 2016; Garza-Rodriguez et al., 2010), Ecuador (García-Vélez et al., 2022). India (Dang & Lanjouw, 2020; Krishna & Shariff, 2011), and South Africa (Schotte et al., 2022). EDE approaches are more robust for measuring poverty dynamics (Sugiharti et al., 2022).

To provide more disaggregate evidence, this study offers poverty components for different groups of individuals to avoid the "one size fits all" indicators often adopted in policymaking. Previous studies in Indonesia have highlighted a higher vulnerability to poverty among people with low levels of education (Taufiq & Dartanto, 2020;), precarious income factors (Noerhidajati et al., 2020), limited access to services (Sugiharti et al., 2022), those located in rural areas (Gibson et al., 2023), and employment in specific sectors (Cameron et al., 2019; Mulyoutami et al., 2020). The literature also points out people vulnerable to poverty based on gender and specific age groups (Mulyoutami et al., 2020). Poverty in Indonesia has decreased at higher rates in provinces that are more open to trade and better connected (Kis-Katos & Sparrow, 2015; Putri et al., 2022) suggesting that accounting for regional differences in poverty measurements is important (Mahadevan et al., 2017).

After the estimation of poverty dynamics for various groups based on socio-demographic and employment characteristics, we apply a quantile approach to examine whether the following set of factors may be associated with poverty in Indonesia (See Table 2 for details):1) demographic characteristics; 2) access to services and government aid programs (education, credit, health insurance, government aid); 3) spatial location (regional, urban-rural); 4) household characteristics; 5) employment; 6) ownership of assets; and 7) negative and positive shocks for households. Additionally, 8) we test whether the most important social aid programs in Indonesia—*Jamkesmas* (health insurance), *Raskin* (subsidized food-rice program for the 30 lowest income groups), *Keluarga Harapan PKH* (social security for the poor), and energy subsidy (gasoline *BBM* and LPG gas)—are linked to chronic and transient poverty alleviation.

This study uses the IFLS from 2007 to 2014, covering 13 out of 27 provinces in Indonesia. We employed a balanced data panel with 12,897 households. Poverty lines per district, city, and period were estimated using the Indonesian Database for Policy and Economic Research (INDO-DAPOER). Previous studies in Indonesia mainly employed poverty lines estimated at the provincial level, primarily distinguishing whether households are located in urban or rural areas. As noted by Akita and Miyata (2018), the inequality between districts within provinces is large, suggesting the need to use disaggregated data to obtain more precise poverty and inequality indicators.

We add to poverty dynamics research by studying Indonesian cases in the following ways. First, we use large longitudinal data to track individuals over time,<sup>1</sup> which is more accurate than synthetic panels that rely on assumptions of income and consumption patterns. Second, we apply the EDE and poverty line at the district level to improve on previous studies, which found transient poverty is higher than chronic poverty. In other developing countries (See Table 1, Mexico, the Philippines, Bangladesh,

Table 2. Desc	riptions of the variables
Variable	Definition (Head of Household—HH)
Edu	The highest education level attained of the head of household
	(1) Not attending school/elementary school (SD)
	(2) Junior high school (SMP)
	(3) High school (SMA) of equivalent
	<ul> <li>(4) Diploma of bachelor degree (S1)</li> <li>(5) Master (S2) or doctoral (S3)</li> </ul>
Gender	Gender of the HH
Age	Age of HH (years)
Age2	Age squared
Employ	Employment status of HH (1 or 0)
Mhh	Number of household members
Spouse_Work	Spouse employment status (1 or 0)
Urban	HH is located in an urban area (1 or 0)
Assets	Value of HH assets (Indonesian Rupiah)
Access	
Finance	Access for all finance types (1 or 0)
Mobility	Access to mobility (ownership of motorized vehicles) (1 or 0)
Cominfo	Access to communication and information (1 or 0)
Shocks	
Shock_Positive	Receives an increase in salary or gifts (1 or 0)
New job	Receives a new job (1 or 0)
Disability	HH has a disability (1 or 0)
Accident	HH has had an accident
Death	HH/member of household has died
Government Pro	ograms
Health	Health assistance and insurance (Jamkesmas) (1 or 0)
Bbm	Access to fuel-gas subsidy (1 or 0)
Pkh	HH Receives social assistance PKH (Keluarga Harapan) (1 or 0)
Raskin	HH Receives basic needs assistance (Raskin) (1 or 0)
Edu	HH Receives assistance with access to education (1 or 0)
Work Sector	
Sector1	Work in mining and agriculture sector (1 or 0)
Sector4	Work in the financial sector
C_Selfemploy	Work as self-employed (have employees or not) (1 or 0)
Casual_Worker	Work casually in agriculture/non/unpaid (1 or 0)

Source. IFLS Data. Note. 1= yes; 0=otherwise. HH Head of Household

South Africa, and Pakistan, among others), poverty dynamics are estimated using more straightforward approaches (spells, components, headcount ratios, or other) and highly aggregated data to determine poverty lines (see Table 1). Third, we examine poverty dynamics in several groups, offering a robust disaggregated analysis. Empirical evidence on the rural—urban scope (Farooq & Ahmad, 2020; Shah & Debnath, 2022), provincial location (Artha & Dartanto, 2018; Tsiboe et al., 2023), and educational gaps (Purwono et al., 2021; Taufiq & Dartanto, 2020) has been extensive. Evidence of disaggregated groups based on gender, sectoral economic activity, job status, and education is limited. Fourth, Indonesia is a large developing nation with 270 million people, 1.9 million km<sup>2</sup>, US\$1.3 trillium in 2022 (GDP), diverse

ethnicity, religious variety, and varied geography. This provides an interesting context for multidimensional poverty research (Borga & D'Ambrosio, 2021; Roy et al., 2019; Shah & Debnath, 2022).

This paper is structured as follows: Section two reviews poverty dynamics. Section three outlines the methodology and data. Section four presents the poverty gap estimation with EDE and quantile regression to examine poverty levels, socioeconomics, and policies. Section five concludes with policy implications and limitations.

### 2. Literature review

## 2.1. Poverty dynamics

The literature on poverty dynamics measurement uses the spell approach, headcount ratio, component approach, among others. The spell approach defines chronically poor people as those in poverty during the entire period (Lo Bue & Palmisano, 2020; Mendola & Busetta, 2012). Those with income or consumption expenditure above the poverty line are transiently poor (Ahmed & Tauseef, 2022). The spell approach does not measure poverty or unequal income distribution (Mai & Mahadevan, 2016). It also does not permit income transfers across periods (Bayudan-Dacuycuy & Lim, 2014).

The component approach, which uses the square poverty gap (Sundar Pani & Mishra, 2022), averages individual income/expenditure, allowing perfect substitutability (Alkire et al., 2017). It assumes a permanent income component, distinguishing chronic poverty from transient poverty. The estimated poverty line is used to determine chronic poverty in households (Shah & Debnath, 2022). The gap between total and chronic poverty indicates transient poverty. However, this approach does not take into account the number of times a household falls below the poverty line (Mai & Mahadevan, 2016).

Building on the components approach, we adopt the EDE method (Jalan & Ravallion, 2000) to relax the assumption of stable income across periods. The EDE adopts the concept of permanent income as the basis for measuring chronic poverty (Muryani & Esquivias, 2021). Permanent income is assumed to be the minimum income/expenditure needed by a household to maintain equivalent welfare levels across periods through inter-temporal income transfers (subject to budget constraints), as noted byHauser (2023).

Mendola and Busetta (2012 developed the Poverty Persistence Index (PPI) and Aggregate Index of Persistence in Poverty (APPI) to measure poverty severity, length, and recentness. The "Incidence, Intensity, Depth, and Severity" studies have studied multidimensional poverty in developing countries (Abubakar, 2022; Roy et al., 2019; Shah & Debnath, 2022), including Indonesia (Firdausy & Budisetyowati, 2022). Synthetic panel data is another way to manage longitudinal data when it is not consistently available. This approach assumes that changes in income and consumption create poverty bounds, which can be used to differentiate the poor from the non-poor (Dang & Dabalen, 2019; Shabnam et al., 2023).

To end poverty in Indonesia, we need to comprehend its chronic and transient causes. The debate on poverty measures in Indonesia is inconclusive. Transient poverty is estimated at 60% or more (Aji, 2015). Dartanto et al. (2020) used EDE and IFLS data to estimate 20% poverty in 2014, twice the official and prior estimates, which appears unlikely. Purwono et al. (2021) estimated poverty with the EDE and spell approaches, with 92% and 29% poverty respectively. Indonesia had especially high EDE poverty, likely due to the poverty lines' high aggregation (Sugiharti et al., 2022). This study seeks to resolve this debate.

## 2.2. Socioeconomic factors associated with changes in poverty

The chronic component of poverty is usually linked to structural factors like lack of education (Tohari et al., 2019), health services, remote/rural locations (Sundar Pani & Mishra, 2022), lack of

credit, and working in agriculture (Ahsan & Kelly, 2018; Cameron et al., 2019; Ruggeri Laderchi et al., 2017; Schaner & Das, 2016). Transient poverty may be caused by sickness, job loss, disasters, price hikes, poor health, low income, and no savings (Dang & Dabalen, 2019; Fitrinitia & Matsuyuki, 2022; Noerhidajati et al., 2020).

Chronic poverty in Indonesia has decreased, with 9.78% (26.4 million) poor in March 2020 (Statistics Indonesia, BPS). However, moderate poverty and economic vulnerability remain high (Firdausy & Budisetyowati, 2022; Gibson et al., 2023). Aji (2015) estimated 25% of the population (65 million people) may be at risk of poverty. Economic shocks, disasters, and the COVID-19 pandemic may raise the poverty rate (Dartanto et al., 2020; Leal Filho et al., 2022). Poor effectiveness of poverty alleviation programs may be due to imprecise identification of chronic and transient poverty, and lack of coordination in Indonesia's programs for the poor and low-income (Bah et al., 2019).

In addition, Indonesia's poverty is more prevalent in rural areas, agricultural activities, and among women (Cameron et al., 2019; Dartanto & Nurkholis, 2013; Mai & Mahadevan, 2016). Smaller and more traditional farmers are also vulnerable, as are those lacking access to insurance, finance, and technology (Firdausy & Budisetyowati, 2022; Muryani & Esquivias, 2021; Sugiharti et al., 2022). Poverty reduction efforts in Indonesia are addressed through three blocks of programs: direct social aid (Tohari et al., 2019), community empowerment, and microenterprise empowerment. Therefore, it is crucial to determine whether policy programs to reduce poverty are effective.

### 3. Methodology

This study estimates chronic and transient poverty components in Indonesia using consumption data. The EDE method proposed by Duclos et al. (2010) measures income gaps among individuals (social welfare and inequality). EDE generates a distribution of poverty gaps (Bayudan-Dacuycuy & Lim, 2013; Mai & Mahadevan, 2016) and distinguishes between chronic and transient components among individuals based on the generated distribution gaps.

For a particular period t, the normalized poverty gap is described as

$$g_{it} = \frac{z_t - y_{it}}{z_t} \tag{1}$$

where  $g_{it}{}^{\alpha} = \left(\frac{z_t - y_{it}}{z_t}\right)^{\alpha}$  if  $z_t > y_{it}$  and  $g_{it}{}^{\alpha} = 0$  if  $z_t \le y_{it}$ 

The normalized gaps across periods of each individual denoted i are by it all individuals, is  $g_i = (g_{i1}, g_{i2}, g_{i3}, \dots, g_{it}, \dots, g_{iT})$ and across given by  $g = (g_1, g_2, g_3, \dots, g_n)$ . Employing a monotonic transformation of  $\Gamma_a$  leads to a measure of total poverty expressed in money terms as

$$\Gamma_{\alpha}^{Total}(g) = \Gamma_{\alpha}(g) = \left[\mathsf{P}_{\alpha}(g)\right]^{\frac{1}{\alpha}}$$
(2)

where  $\Gamma_{\alpha}(g)$  is defined as the EDE poverty gap. To account for the inequality in poverty status,  $\alpha$  is set to be  $\geq 1$ , meaning that the larger the difference between  $\Gamma_{\alpha}(g)$  and  $\Gamma_{1}(g)$ , the more unequal the distribution of individual welfare is (ill-fare) and the more normalized poverty gaps are. The cost of inequality in the gaps among the entire population is set by

$$C_{\alpha}(g) = \Gamma_{\alpha}(g) - \Gamma_{1}(g)$$
(3)

where  $C_{\alpha}(g)$  is the cost of an increase in the average poverty gap, expressed in monetary terms. The total poverty is given by equations (2) and (3) as

$$\Gamma_{\alpha}^{Total}(g) = \Gamma_{\alpha}(g) = \mathsf{C}_{\alpha}(g) + \Gamma_{1}(g) \tag{4}$$

The component of transient poverty is obtained by

$$\theta_a(\mathbf{g}_i) = \gamma_a(\mathbf{g}_i) - \gamma_1(\mathbf{g}_i) \tag{5}$$

Aggregating this transient cost across the entire population of individuals (n's) gives

$$\Gamma_{\alpha}^{Transient}(g) = \frac{1}{n} \sum_{i=1}^{n} \theta_{\alpha}(g_i)$$
(6)

Following Duclos et al. (2010), the distribution of individual EDE poverty gaps are defined as  $\gamma_{\alpha}(g_i)$ , while the cost of inequality between individuals as  $C_{\alpha}(\gamma_{\alpha}(g_i))$  and the transient poverty as

$$\Gamma_{\alpha}^{\text{TotalPoverty}}(\boldsymbol{g}) = \Gamma_{\alpha}^{\text{Transient}}(\boldsymbol{g}) + \Gamma_{\alpha}^{\text{Chronic}}(\boldsymbol{g})$$
(7)

where

$$\Gamma_{\alpha}^{Chronic}(g) = \Gamma_{1}(g) + C_{\alpha}(\gamma_{\alpha}(g_{i}))$$
(8)

#### 3.1. Quantile regression approach

This study computed poverty gaps and used quantile and OLS regression to analyse chronic and transient poverty determinants. Koenker and Bassett (1978) quantile regression model can assess correlated effects of variables in different response variable quantiles. Quantile regressions enable non-constant parameters across welfare spectrums (De Silva, 2008). Quantile analysis can assess the impact of socioeconomic and policy factors on poverty levels. This approach compares the effects of poverty-related factors at different welfare levels (Peng et al., 2019). Following Bayudan-Dacuycuy and Lim (2013), the model is expressed as:

$$Y_{i} = BX_{i} + e_{i},$$

$$Quant_{\theta}(y_{i}|x_{i}) = \beta_{\theta}x_{i}$$

$$Y_{i}* = 0, if Y_{1} = 0$$

$$Y_{i}* = Y_{i} if 0 < Y_{i} \le 1$$
(9)

where  $Y_i^*$  is the dependent variable identifying the limit between chronic and transient poverty, represented as the poverty status;  $X_i$  is the vector of variables employed as explanatory components; B is the vector of unknown socioeconomic parameters to be estimated;  $e_i$  is a disturbance term assumed to be independent, following a normal distribution, with a mean of zero and a constant variance. I = 1, 2, ... (*n* total observations).

We estimate three quantile regressions at the 25, 50, and 75 quantiles. Standard errors were estimated by bootstrapping with 100 replications following De Silva (2008). As a point of comparison, we provide estimates using OLS, which allows an estimation of the relation between the independent set of variables and the average value of the response variable (Garza-Rodriguez et al., 2021). Meanwhile, quantile regressions enable the estimation of the effects of the explanatory variable on different wealth levels.

The independent variable for both the chronic and transient models considers the following aspects (see Table 2): 1) demographic characteristics; 2) household characteristics; 3) spatial location; 4) access to services; 5) the presence of positive shocks and negative socks; 6) social

programs (*Jamkesmas*, health insurance), energy subsidy, *Keluarga Harapan PKH* (social security for the poor), *Raskin* (subsidized rice program), and public education; and 7) employment.

#### 3.2. Data

Data were collected from the Rand Corporation (IFLS). We focus on the period between 2007 and 2014. The rationale for selecting this period is that we draw poverty lines at the district level, and previous periods do not have such data. The IFLS is longitudinal data consisting of 13 of the 27 provinces in Indonesia, representing 83% of the country's population. This study used household-level data in the form of a balanced panel with 25,794 observations. We employ the poverty lines per district, city, and period provided by INDO-DAPOER.

To measure poverty, we employed data on household expenditure instead of income, as it is a more consistent proxy for household welfare in developing countries (Bayudan-Dacuycuy & Lim, 2013; Mai & Mahadevan, 2016). Household expenditure is adjusted according to the number of household members, employing the equivalence scale from (Glewwe & Twum-Baah, 1991).  $AE = (N_{adults} + \beta N_{children})^{\mu}$  where  $\mu \leq 1$  works as a parameter to capture the scale effects, N accounts for the number of household members, and  $\beta$  adjusts for a child's cost (adjusted at different age groups) relative to an adult's cost.

Due to space limitations, the descriptive statistics of the observational data are not displayed (available upon request).

#### 4. Results

#### 4.1. Poverty dynamics

Table 3 displays the total and partial poverty estimates for each category. Systematically, Indonesia's largest proportion of dynamic poverty is chronic, with significant variations across groups. A total of 76.6% of poverty is chronic, almost triple that of transient or temporary poverty (23.3%). Sumatra Island provinces have the highest average poverty rate (11%). West Nusa Tenggara has 10% poverty, with 61% being chronic. South Sumatra (85%), West Nusa Tenggara (83%), and South Sulawesi (88%) have the highest chronic poverty proportions. South Sulawesi had the highest chronic poverty rate (93%). Riau, Jambi, Central Kalimantan, and East Kalimantan had the highest cost of inequality (42–52%). Sumatra and Kalimantan are rich in natural resources, such as palm oil, rubber, and coal. Inequality's cost implies substantial inequity in both islands, despite resources. This may be due to Dutch Disease.

Our EDE-based estimates of total, chronic, and transient poverty in Indonesia differ from non-EDE studies, which usually found a larger proportion of transient poverty (Akita & Dariwardani, 2013; Dartanto & Nurkholis, 2013). However, our estimates match those of other EDE-based studies in Indonesia (Mai and Mahadevan (2016). They found 76% chronic poverty in Indonesia between 1993–2007. Dartanto et al. (2020) used IFLS data and EDE, yielding different estimates due to different poverty lines. Sugiharti et al. (2022) found chronic poverty as that the largest component in Indonesia.

This study, using 2007–2014 data, finds lower inequality costs and higher average poverty gaps than Mai and Mahadevan (2016). Previous research assumed equal poverty thresholds across regencies and cities in a province, which is doubtful given the significant income/expenditure variations (Firdausy & Budisetyowati, 2022; Sugiharti et al., 2022). Mai and Mahadevan (2016) used an individual-scale and provincial poverty line, whereas we used household-scale data and a more detailed poverty line at district and city levels.

Poverty decreased overall. Table 3 reveals disparities among households. Household heads with only elementary school or less make up 16% of total poverty and 82% of chronic poverty. Education attainment reduces total and chronic poverty, increasing transient poverty and inequality cost. Total poverty among those with senior high school dropped to 4.5%, and chronic poverty decreased from

Table 3. Pover	ty decompositio	n (total, transie	nt, and chronic)		
				Decompositio pov	on of chronic erty
	Total poverty	Transient Poverty	Chronic poverty	Average poverty	Cost of Inequality
By Education					
Elementary or less	0.160	0.028(18%)	0.132(82%)	0.089(56%)	0.042(26%)
Junior High school	0.059	0.014(24%)	0.044(76%)	0.032(56%)	0.012(21%)
Senior High school	0.045	0.013(29%)	0.032(71%)	0.020(47%)	0.011(26%)
Undergraduate	0.025	0.009(37%)	0.015(63%)	0.006(27%)	0.008(37%)
Postgraduate	0.019	0.008(43%)	0.010(57%)	0.004(23%)	0.006(33%)
By Province					
North Sumatra	0.118	0.025(22%)	0.092(78%)	0.053(45%)	0.039(33%)
West Sumatra	0.100	0.023(23%)	0.077(77%)	0.045(44%)	0.032(31%)
Riau	0.089	0.015(18%)	0.073(87%)	0.060(11%)	0.012(22%)
Jambi	0.021	0.010(50%)	0.010(50%)	0(0%)	0.010(49%)
South Sumatra	0.122	0.023(19%)	0.098(81%)	0.065(56%)	0.033(28%)
Lampung	0.093	0.020(22%)	0.072(78%)	0.051(54%)	0.021(22%)
Bangka Belitung	0.107	0.024(23%)	0.083(77%)	0.056(52%)	0.026(24%)
Riau Islands	0.039	0.012(31%)	0.027(69%)	0.006(16%)	0.021(52%)
Jakarta	0.068	0.016(24%)	0.051(76%)	0.034(50%)	0.017(24%)
West Java	0.082	0.018(22%)	0.064(78%)	0.045(56%)	0.019(23%)
Central Java	0.059	0.013(23%)	0.045(77%)	0.036(61%)	0.009(15%)
Yogyakarta	0.056	0.013(24%)	0.042(76%)	0.028(51%)	0.013(24%)
East Java	0.063	0.015(24%)	0.048(76%)	0.032(51%)	0.015(24%)
Banten	0.052	0.016(31%)	0.036(69%)	0.023(43%)	0.013(25%)
Bali	0.022	0.005(26%)	0.017(74%)	0.012(52%)	0.004(21%)
West Nusa Tenggara	0.105	0.018(17%)	0.087(83%)	0.064(61%)	0.022(21%)
Central Kalimantan	0.036	0.010(29%)	0.026(71%)	0.010(28%)	0.015(42%)
South Kalimantan	0.055	0.008(15%)	0.046(85%)	0.036(65%)	0.010(19%)
East Kalimantan	0.006	0.003(49%)	0.003(49%)	0(0%)	0.003(49%)
South Sulawesi	0.069	0.008(12%)	0.060(88%)	0.053(93%)	0.006(12%)
By Gender					
Male	0.067	0.015(23%)	0.051(77%)	0.035(54%)	0.015(23%)
Female	0.116	0.023(20%)	0.093(80%)	0.062(54%)	0.031(27%)
By Marital Status					
Single	0.115	0.022(20%)	0.092(80%)	0.059(52%)	0.032(28%)
Married	0.065	0.015(23%)	0.050(77%)	0.035(55%)	0.014(23%)
By Employment Status					
No-Job	0.116	0.023(20%)	0.092(80%)	0.061(53%)	0.031(27%)
Employed	0.069	0.015(23%)	0.053(77%)	0.037(54%)	0.016(24%)

(Continued)

				Decompositi pov	on of chronic erty
	Total poverty	Transient Poverty	Chronic poverty	Average poverty	Cost of Inequality
By Location					
Rural	0.092	0.018(20%)	0.073(80%)	0.052(57%)	0.021(23%)
Urban	0.063	0.015(24%)	0.048(76%)	0.032(51%)	0.016(26%)
By Farming					
No-Farming	0.069	0.016(23%)	0.053(77%)	0.035(52%)	0.017(25%)
Farming	0.089	0.018(21%)	0.070(79%)	0.050(57%)	0.020(22%)
By Job Status					
Self Employed	0.084	0.017(21%)	0.066(79%)	0.047(56%)	0.019(23%)
Gov Worker	0.025	0.009(37%)	0.016(63%)	0.008(35%)	0.007(30%)
Casual Unpaid	0.094	0.020(22%)	0.073(78%)	0.047(50%)	0.026(28%)
Casual Agriculture	0.126	0.022(18%)	0.103(82%)	0.073(58%)	0.029(23%)
Casual Non_Agriculture	0.083	0.018(22%)	0.064(78%)	0.044(54%)	0.020(24%)
By Sector					
Mining	0.093	0.018(20%)	0.074(80%)	0.049(58%)	0.025(29%)
Manufacturing	0.056	0.014(26%)	0.041(74%)	0.027(51%)	0.013(25%)
Electricity'	0.053	0.016(30%)	0.037(70%)	0.017(33%)	0.019(36%)
Construction	0.061	0.015(25%)	0.046(75%)	0.030(50%)	0.015(26%)
Wholesale	0.054	0.013(24%)	0.040(75%)	0.029(54%)	0.011(21%)
Transportation	0.059	0.013(23%)	0.045(77%)	0.032(56%)	0.013(23%)
Finance	0.040	0.013(33%)	0.027(67%)	0.012(31%)	0.014(35%)
Services	0.050	0.013(26%)	0.037(74%)	0.024(49%)	0.013(26%)
Other	0.076	0.019(25%)	0.057(75%)	0.036(48%)	0.021(28%)
Agriculture	0.107	0.020(19%)	0.086(81%)	0.062(58%)	0.023(22%)
Farmer	0.089	0.018(21%)	0.070(79%)	0.050(57%)	0.020(22%)

Source: Calculations based on IFLS Data for 2007 and 2014.

82% to 71%. We demonstrate how education can move people from chronic to transient poverty, in line with earlier studies (Dartanto et al., 2020; Taufiq & Dartanto, 2020).

Females experience almost twice the total poverty of males and a higher chronic poverty rate. Muryani and Esquivias (2021) attributed this to education, financial access, and location. Single heads of households had a higher total poverty rate (11.5%) than married households. Single households experience more income inequality (28%) than married couples (23%). Unemployed rural farming households have higher poverty and chronic poverty than urban households. The rural-urban and agricultural/non-agricultural gaps are narrower than other groups, similar to the cost of inequality. Moeis et al. (2020) found that welfare improved (2000–2007) when Indonesians left agriculture. However, casual agriculture workers had the highest average total poverty (12.6%) and chronic/average poverty based on job status (8.3%).

The total poverty among self-employed heads of households is 8.4%, lower than average. Dartanto et al. (2020) found similar results, noting informal workers are unlikely to move into higher-income groups. Entrepreneurship has been encouraged (Cameron et al., 2019) as informal workers and self-

employed people make up 60–70% of the workforce (Muryani & Esquivias, 2021; Sugiharti et al., 2022). Urban women are often in informal sectors (Schaner & Das, 2016). Government workers have the least poverty but the highest inequality and poverty fluctuation among job groups. Mining and agriculture have the highest chronic and average poverty (in line with Moeis et al., 2020), while electricity and finance have the highest inequality costs. Manufacturing and service-related industries have 40–50% lower poverty than primary sectors.

### 4.2. Quantile regression estimates

This section empirically estimates the effects of demographic, socioeconomic, and policy aspects on the dynamic poverty components using quantile regression (Table 4). OLS estimations were used to compare results. Demographic and social factors were linked to chronic poverty (Table 4). Education, gender, age, employment, household size, spouse's employment, and location all had positive correlations with lower chronic poverty. These variables had greater effects at higher quantiles. Fewer variables were associated with transient poverty, showing socio-demographic characteristics are important for chronic poverty but less so for temporary poverty.

Education levels of household heads reduce chronic poverty, but not transient poverty. Education is more strongly linked to chronic poverty in the upper quartiles. This is consistent with previous research in Indonesia (Absor et al., 2022; Dartanto & Nurkholis, 2013; De Silva & Sumarto, 2015; Hanandita & Tampubolon, 2016) and other developing countries (Janz et al., 2023; Leal Filho et al., 2022; Nawab et al., 2023; Shah & Debnath, 2022).

Female-headed households in Indonesia experience more poverty (11.5%) than male-headed ones (6%). Prior research indicates women are more likely to be poor or earn less than men (Bella & Dartanto, 2018; Muryani & Esquivias, 2021). Gender poverty gaps persist as a significant issue. Education seems to play a more significant role than gender in poverty, consistent with studies in Indonesia, India, Kenya, and Mexico (Absor et al., 2022; Abubakar, 2022; Dartanto et al., 2020; Garza-Rodriguez et al., 2021; Ichwara et al., 2023; Shah & Debnath, 2022). The age of the house-hold head is also linked to poverty in a non-linear manner. Older household heads are more likely to experience poverty, aligning with previous studies (Peng et al., 2019).

OLS calculations suggest employment reduces chronic poverty by 1.7% and temporary poverty by 0.4%. However, employment in primary sectors (Work\_Sector1) doubles chronic poverty risk compared to other sectors. This indicates that merely having a job doesn't guarantee poverty escape, and primary sector households are more poverty-prone. These findings align with previous studies in Indonesia, Nigeria and Mexico, where working in secondary and tertiary sectors is linked to reduced poverty levels (Abubakar, 2022; Garza-Rodriguez et al., 2021).

Family size increases the risk of total poverty by 2%. Chronic poverty (1.9%) is more likely to increase than transient poverty (0.1%), especially in the upper quartiles. More family members in poor households may increase the chance of staying poor rather than escaping poverty. These findings align with earlier studies in Indonesia (Artha & Dartanto, 2018; Bella & Dartanto, 2018; Muryani & Esquivias, 2021; Widyanti et al., 2009), the Philippines (Bayudan-Dacuycuy & Lim, 2014), Sri Lanka (Deyshappriya & Minuwanthi, 2020), Mexico (Garza-Rodriguez et al., 2021), and Ethiopia (Mehari, 2022), although the quartile tests were not provided in earlier analyses in Indonesia.

45% of households have two incomes, and having employed spouses reduces poverty. The effect of a partner's employment status is greater in deeper poverty quintiles. Indonesian women are often out of the labor force (Cameron et al., 2019; Schaner & Das, 2016), so policies to encourage women's participation may reduce poverty. Improving childcare and introducing flexible, family-friendly arrangements can help more women join the workforce (Peng et al., 2019).

Living in urban areas negatively correlates with chronic and total poverty, aligning with (Tsiboe et al., 2023), but not with transient poverty. Previous research indicates higher poverty rates and

Table 4. Determina	nts of pover	ty by demog	graphics, acc	:ess, shocks	, and Gover	nment prog	ams (trans	ient, chroni	c, & total)			
		TRANSIE	NT POOR			CHRONI	C POOR			TOTAL P	OVERTY	
	OLS	Q25	Q50	Q75	OLS	Q25	Q50	Q75	OLS	Q25	Q50	Q75
Edu	-0.001	-0.001	0.000	-0.003	-0.037***	-0.0083***	-0.024***	-0.050***	-0.039***	-0.011***	-0.028***	-0.050***
Sex	0.000	0.001	0.001	0.000	0.008*	0.002	0.006*	0.023***	0.008	0.003	0.005	0.024***
Age	0.0001*	0.000	0.000	0.000	0.001***	0.0003***	0.000***	0.002***	0.001***	0.000	0.001	0.002
Age2	-0.000**	0.000	0.000	0.000	-0.000***	-0.000***	-0.000***	-0.000***	-0.000***	0.000	0.000	0.000
Employ	-0.004*	-0.002	0.000	-0.006	-0.017*	-0.007	-0.015*	-0.012	-0.021*	-0.013	-0.014	-0.018
МҺҺ	0.001***	0.000***	0.001**	0.001***	0.019***	0.005***	0.014***	0.027***	0.020***	0.007***	0.016***	0.029***
Spouse_Work	-0.002*	-0.001	-0.002	-0.005*	-0.029***	-0.006***	-0.013***	-0.050***	-0.032***	-0.010*	-0.023***	-0.049***
Urban	-0.001	-0.002	0.000	-0.001	-0.018**	-0.002	-0.003	-0.025**	-0.019**	-0.005	-0.005	-0.020
Assets	0.000	0.000	-9.12e-12*	0.000	-1.02e- 10***	-2.31e- 11***	-5.10e- 11***	-6.99e- 11***	-1.07e- 10***	-3.29e- 11***	-6.29e- 11***	-7.07e- 11***
Access												
Finance	-0.009***	-0.012**	-0.014**	-0.007	-0.123***	-0.056**	-0.176***	$-0.171^{***}$	-0.132***	-0.082***	-0.164***	-0.204***
Mobility	-0.004**	-0.001	-0.004	-0.008***	-0.059***	-0.012***	-0.036***	-0.085***	-0.063***	-0.016**	-0.037***	-0.093***
Cominfo	-0.009***	-0.007**	-0.012**	-0.014***	$-0.111^{***}$	-0.032***	-0.113***	-0.212***	-0.120***	-0.042***	-0.116***	-0.213***
Shocks												
Shock_Positive	-0.005***	-0.004***	-0.006**	-0.006**	-0.031***	-0.009***	-0.024***	-0.027**	-0.037***	-0.012**	-0.032***	-0.037***
New job	-0.005	-0.007**	-0.009*	-0.003	0.000	-0.003	-0.007	-0.006	-0.005	-0.007	-0.014	-0.008
Disability	-0.007*	0.013	-0.010*	-0.011	0.041**	0.047**	-0.003	0.042*	0.035	0.076**	0.003	0.006
Accident	0.001	0.006*	-0.001	0.001	-0.003	-0.004	-0.001	0.006	-0.002	-0.010	-0.001	0.009
Death	0.023**	0.025**	0.024	0.022	0.066	0.066*	0.074***	0.066	*060.0	0.116***	0.094**	0.016
Government Programs												
Health	-0.001	-0.001	-0.003	-0.003	-0.046***	-0.009***	-0.025***	-0.047***	-0.047***	$-0.011^{**}$	-0.029***	-0.050***
Bbm	0.000	0.003*	-0.001	-0.002	0.067***	0.020**	0.071***	0.129***	0.068***	0.035***	0.070***	0.129***
Pkh	-0.002	-0.003	-0.011	0.003	-0.061**	-0.036	-0.012	-0.060	-0.063*	-0.059	0.000	-0.068
												(Continued)

Sugiharti et al., Cogent Economics & Finance (2023), 11: 2267927 https://doi.org/10.1080/23322039.2023.2267927

Table 4. (Continued)												
		TRANSIE	NT POOR			CHRONI	C POOR			TOTAL P	OVERTY	
	OLS	Q25	Q50	Q75	OLS	Q25	Q50	Q75	OLS	Q25	Q50	Q75
Raskin	-0.001**	-0.001***	-0.002***	-0.002	0.002	-0.001***	-0.002*	0.001	0.000	-0.003***	-0.005***	0.001
Edu	-0.003	-0.003	-0.002	0.000	-0.040**	-0.012	-0.024**	-0.047***	-0.043**	-0.021*	-0.036***	-0.072***
Work—Sector												
Sector1	0.004*	0.004*	0.001	0.008	0.041***	0.023***	0.043***	0.034	0.045***	0.037***	0.043***	0.034*
Sector4	0.012**	0.001	0.014***	0.014*	0.019	0.016*	0.023***	-0.023	0:030	0.026**	0.035*	0.005
C_Selfemploy	-0.004**	0.000	-0.005	-0.007*	-0.013	-0.007	-0.011**	-0.015**	-0.016*	-0.009	-0.015	-0.015
Casual_Worker	0.001	0.000	0.005	0.001	0.023*	-0.002	-0.004	0.053**	0.024*	-0.005	0.002	0.049**
Source: Calculations base	ed on IFLS Date	a for 2007 and	2014. Note. **'	*, ** and * denc	ote significance	e at the 1%, 5%	6 and 10% lev€	sls, respectively				

intensity in rural Indonesia compared to urban areas (Dartanto et al., 2020; Hanandita & Tampubolon, 2016; Mai & Mahadevan, 2016). Reports from Morocco (Yassine & Bakass, 2022), Mexico (Fernández-Ramos et al., 2016), South Africa (Biyase & Zwane, 2018), and Sri Lanka (Deyshappriya & Minuwanthi, 2020). However, we posit that rural poverty is persistent.

Asset ownership reduces total, chronic, and transient poverty, as seen in India (Khosla et al., 2023), the Philippines (Bayudan-Dacuycuy & Lim, 2013), and Indonesia (Dartanto et al., 2020). Assisting individuals in acquiring assets like land or houses could help alleviate poverty, in line with Peng et al. (2019). Access to finance, mobility, info, and communication reduces poverty, especially chronic poverty. Infrastructure (Gibson et al., 2023)), mobility, telecommunications, and financial services (Esquivias et al., 2020) are essential for poverty alleviation. Credit exclusion is often linked to chronic poverty (Esquivias et al., 2020; Ruggeri Laderchi et al., 2017). Households' capacity to cope with shocks affects their risk of temporary poverty. Positive shocks like bonuses, salary increases, and gifts reduce poverty (both chronic and transient). New jobs can reduce temporary poverty, mainly in Q25 and Q50, but not chronic poverty. A new job can stabilise income and reduce the risk of poverty.

Negative events like accidents are linked to short-term poverty. The death of family heads or relatives raises the chance of chronic (6% at Q25, 7.4% at Q50) and transient poverty (2.5% at Q25, 2.3% at OLS), necessitating social aid due to long-term effects. Short-term effects could intensify after a family leader's death, which is significant for COVID-19 (Sparrow et al., 2020). Increased mortality may affect poverty in Indonesia and globally (Albani et al., 2022; Brandily et al., 2021; Laajaj et al., 2022). Having a disabled household member raises chronic poverty odds, as Peng et al. (2019) found. Surprisingly, this study discovered households with disabilities in the Q50 are less likely to experience temporary poverty. Poor health (Khosla et al., 2023; Özsoy & Gürler, 2022), disabilities (Absor et al., 2022), and accidents (Ahsan & Kelly, 2018; Bella & Dartanto, 2018) can negatively impact income or assets due to inability to work, asset utilization, or job market discrimination (Usman & Projo, 2021).

Government aid like health insurance and scholarships substantially lessen chronic poverty, as shown in Indonesian studies (Hanandita & Tampubolon, 2016; Moeis et al., 2020). Health assistance and public education reduce chronic poverty by 4.6% and 4% respectively, according to OLS estimates. Support programs are crucial in mitigating chronic poverty (Bah et al., 2019; Tohari et al., 2019). Our study aligns with Sparrow et al. (2013), suggesting health insurance reduces poverty. Food aid is negatively linked to both persistent and temporary poverty (0.1–0.2%). A substantial aid program reduced chronic and total poverty based on OLS results, but quantile estimates were insignificant.

Fuel subsidies, aimed at universally boosting Indonesians' purchasing power, paradoxically increase temporary and chronic poverty risk (Sahara et al., 2022). Non-poor households received 74.5% of these subsidies, leaving only 25.5% for poor households, indicating a skewed distribution.

In Indonesia, the job sector of household heads significantly affects poverty dynamics. Individuals employed in mining, agriculture, and finance sectors are more prone to chronic and temporary poverty due to wage differences (Gibson et al., 2023; Moeis et al., 2020). Non-productivity factors cause disparities (Ahsan & Kelly, 2018; Dartanto et al., 2018). Self-employers face less chronic or transient poverty, as Moeis et al. (2020) found that moving to self-employment or formal activities increases spending and reduces poverty in Indonesia as well as (Chung & Maguire-Jack, 2020) found in the United States. However, Sugiharti et al. (2022) noted a decrease in poverty reduction via self-employment. Casual workers are more likely to face severe, long-term poverty. Q75% households with casual workers had a 5% higher poverty likelihood, according to Sugiharti et al. (2022). Casual workers in Indonesia, often in seasonal jobs, had the highest occupational mobility.

# 5. Conclusion

The estimates of poverty components using the EDE approach indicate that chronic poverty is more prevalent in Indonesia (more than 75%) than transient share. The EDE approach estimates

chronic poverty is more common in Indonesia (over 75%) than transient poverty. Low education, female gender, single mothers, informal jobs, and primary sector work are associated with higher poverty levels. Socioeconomic groups, location, and labor status have large disparities, suggesting targeted efforts for vulnerable groups. Inequality costs more for those with higher education and those in the service sector (finance, energy, and government) or entrepreneurs.

Chronic poverty is higher in most groups, so poverty reduction needs structural policies to improve human resources through education, access to services (health, education, transportation, communication, and finance), and formal employment. Poverty is more severe in areas with natural resources, likely due to Dutch Disease from commodity price fluctuations. Higher education, urban living, larger assets, and formal employment reduced chronic poverty. Access to finance, mobility, communication, and information lessened transient and chronic poverty. Government programs such as health services, food support, direct transfers, and education also reduced chronic poverty. Energy subsidies don't reduce poverty, but may support purchasing power. Disabilities, accidents, and deaths increase chronic and transient poverty.

This study has several drawbacks that should be addressed in future research. It only uses two IFLS surveys from 2007–2014, so more recent data should be included when available. Before 2007, district-level poverty line data was unavailable. Future studies may use alternative methods to estimate the poverty line at the district (or lower) level, extending the analysis period. Despite the data limitation, we can demonstrate the importance of using more detailed data for the poverty line. The study found that inequality is the main cause of persistent poverty in education groups, regions with abundant resources, and certain sectors (e.g. electricity and finance). This implies large income and wage gaps due to higher education or certain sectors (rich in natural resources). Further research is needed to understand the causes of these large disparities. Third, the data and methods did not allow causal inferences on monetary poverty changes over time. The EDE approach offers chronic and transient poverty measures, but not individual household poverty measures (Mai & Mahadevan, 2016). However, we provide chronic poverty sources at a disaggregated level for various categories, which is beneficial for policymakers. The quantile approach cannot prove cause-and-effect between socioeconomic, policy, or life event factors and poverty. However, it reveals the different relationships these elements have across poverty levels.

Our findings point to five research directions. First, use more detailed data to measure poverty due to its sensitivity to living standards, poverty lines, and equivalence scale. Second, poverty reduction efforts must be tailored to each context. Future studies should measure poverty accurately and examine poverty reduction in specific groups. Indonesia should focus on reducing chronic poverty, avoiding transitory poverty, and promoting equal income growth. Studies should assess policy effectiveness and focus on specific reduction programs. Fourth, social programs reduce poverty. Further research should assess direct social benefits and other programs (e.g., conditional transfer schemes) in the country. Non-direct social benefit schemes should be studied to promote entrepreneurship and women's labor market involvement. Childcare and pre-employment cards (Kartu Prakerja) may reduce poverty. Future studies should assess if social aid coordination and common targeting programs can help. Our study could not determine if vulnerable households were reached.

#### Acknowledgments

This work was supported by Research Grant Hibah Penelitian Tahun 2020 of the Universitas Airlangga. APC was supported by Universitas Airlangga, Surabaya, Indonesia.

#### Funding

The work was supported by the Universitas Airlangga  $\left[\mathrm{NA}\right].$ 

#### Author details

Lilik Sugiharti<sup>1</sup> ORCID ID: http://orcid.org/0000-0001-5156-7929 Miguel Angel Esquivias<sup>1</sup> E-mail: miguel@feb.unair.ac.id ORCID ID: http://orcid.org/0000-0002-1282-6163 Mohd Shahidan Shaari<sup>2</sup> Ari Dwi Jayanti<sup>1</sup> ORCID ID: http://orcid.org/0000-0001-9153-5702

Abdul Rahim Ridzuan<sup>3,4,5,6,7</sup>

- <sup>1</sup> Faculty of Economics and Business, Universitas Airlangga, Surabaya, Indonesia.
- <sup>2</sup> Faculty of Business & Communication, Universiti Malaysia Perlis, Arau, Perlis, Malaysia.
- <sup>3</sup> Faculty of Business and Management, Universiti Teknologi MARA, Melaka, Malaysia.
- <sup>4</sup> Institute for Big Data Analytics and Artificial Intelligence (IBDAAI), Universiti Teknologi MARA Shah Alam, Malaysia.
- <sup>5</sup> Centre for Economic Development and Policy, Universiti Malaysia Sabah, Kota Kinabalu, Sabah, Malaysia.

- <sup>6</sup> Institute for Research on Socio Economic Policy, Universiti Teknologi MARA, Melaka, Malaysia.
   <sup>7</sup> Accounting Research Institute (ARI), Universiti Teknologi
- MARA, Shah Alam, Malaysia.

#### **Disclosure statement**

No potential conflict of interest was reported by the author(s).

#### Data availability statement

Data were obtained from the Indonesian Family Life Survey (IFLS). https://www.rand.org/well-being/social-and -behavioral-policy/data/FLS/IFLS.html

#### **Citation information**

Cite this article as: Indonesia's poverty puzzle: Chronic vs. transient poverty dynamics, Lilik Sugiharti, Miguel Angel Esquivias, Mohd Shahidan Shaari, Ari Dwi Jayanti & Abdul Rahim Ridzuan, *Cogent Economics & Finance* (2023), 11: 2267927.

#### Note

1. We only use two out of the five available datasets in Indonesia. We provide new estimates for poverty lines for the 2007–2014 period (two latest datasets). Due to data limitations, the estimation of poverty lines at the district level before 2007 is not possible.

#### References

- Abanokova, K., Dang, H. H., & Lokshin, M. (2022). Do adjustments for equivalence scales affect poverty dynamics? Evidence from the Russian federation during 1994–2017. *Review of Income and Wealth*, 68 (S1). https://doi.org/10.1111/roiw.12559
- Absor, M. U., McDonald, P., & Utomo, A. (2022). Economic disadvantage among older people in rural Indonesia: Risk and protective factors. *Journal of Population Ageing*. https://doi.org/10.1007/s12062-022-09406-1
- Abubakar, I. R. (2022). Multidimensional poverty among Nigerian households: Sustainable development implications. Social Indicators Research, 164(2), 993–1014. https://doi.org/10.1007/s11205-022-02963-0
- Aginta, H., Gunawan, A. B., & Mendez, C. (2021). Regional income disparities and convergence clubs in Indonesia: New district-level evidence. *Journal of the Asia Pacific Economy*, 28(1), 101–132. https://doi.org/ 10.1080/13547860.2020.1868107
- Ahmed, A. U., & Tauseef, S. (2022). Climbing up the ladder and watching out for the fall: Poverty dynamics in rural Bangladesh. Social Indicators Research, 160(1), 309–340. https://doi.org/10.1007/s11205-021-02808-2
- Ahsan, M. N., & Kelly, I. R. (2018). Earnings gaps for conspicuous characteristics: Evidence from Indonesia: Earnings gaps for conspicuous characteristics. Southern Economic Journal, 85(1), 121–141. https:// doi.org/10.1002/soej.12298
- Aikaeli, J., Garcés-Urzainqui, D., & Mdadila, K. (2021). Understanding poverty dynamics and vulnerability in Tanzania: 2012–2018. Review of Development Economics, 25(4), 1869–1894. https://doi.org/10. 1111/rode.12829
- Aji, P. (2015). Summary of Indonesia's poverty analysis. ADB Papers on Indonesia, 4.
- Akita, T., & Dariwardani, N. M. I. (2013). Chronic and transient poverty in Indonesia: A spatial perspective with the 2008-2010 Susenas panel data. *Economics & Management Series*, 2013, 1–26. https://iuj.repo.nii. ac.jp/records/504
- Akita, T., & Miyata, S. (2018). Spatial inequalities in Indonesia, 1996–2010: A hierarchical decomposition

analysis. Social Indicators Research, 138(3), 829–852. https://doi.org/10.1007/s11205-017-1694-1

- Albani, V., Welsh, C. E., Brown, H., Matthews, F. E., & Bambra, C. (2022). Explaining the deprivation gap in COVID-19 mortality rates: A decomposition analysis of geographical inequalities in England. *Social Science & Medicine*, 311, 115319. https://doi.org/10. 1016/j.socscimed.2022.115319
- Alkire, S., Apablaza, M., Chakravarty, S., & Yalonetzky, G. (2017). Measuring chronic multidimensional poverty. *Journal of Policy Modeling*, 39(6), 983–1006. https:// doi.org/10.1016/j.jpolmod.2017.05.020
- Arif, G. M., Iqbal, N., & Farooq, S. (2011). The persistence and transition of rural poverty in Pakistan: 1998-2004. Working Papers & Research Reports, 2011. https://core.ac.uk/download/pdf/6489619.pdf
- Artha, D. R. P., & Dartanto, T. (2018). The multidimensional approach to poverty measurement in Indonesia: Measurements, determinants and its policy implications. Journal of Economic Cooperation and Development, 39(3), 1–38.
- Bah, A., Bazzi, S., Sumarto, S., & Tobias, J. (2019). Finding the poor vs. measuring their poverty: Exploring the drivers of targeting effectiveness in Indonesia. *The World Bank Economic Review*, 33(3), 573–597. https:// doi.org/10.1093/wber/lhx020
- Bayudan-Dacuycuy, C., & Lim, J. A. (2013). Family size, household shocks and chronic and transient poverty in the Philippines. *Journal of Asian Economics*, 29, 101–112. https://doi.org/10.1016/j.asieco.2013.10.001
- Bayudan-Dacuycuy, C., & Lim, J. A. (2014). Chronic and transient poverty and vulnerability to poverty in the Philippines: Evidence using a simple spells approach. Social Indicators Research, 118(1), 389–413. https:// doi.org/10.1007/s11205-013-0409-5
- Beegle, K., Christiaensen, L., Dabalen, A., & Gaddis, I. (2016). Poverty in a rising Africa. World Bank Publications.
- Bella, A., & Dartanto, T. (2018). Persons With Disabilities (PWD) and poverty in Indonesia. Malaysian Journal of Economic Studies, 55(2), 167–188. https://doi.org/10. 22452/MJES.vol55no2.2
- Biyase, M., & Zwane, T. (2018). An empirical analysis of the determinants of poverty and household welfare in South Africa. *The Journal of Developing Areas*, 52 (1), 115–130. https://doi.org/10.1353/jda.2018.0008
- Borga, L. G., & D'Ambrosio, C. (2021). Social protection and multidimensional poverty: Lessons from Ethiopia, India and Peru. World Development, 147, 105634. https://doi.org/10.1016/j.worlddev.2021. 105634
- Brandily, P., Brébion, C., Briole, S., & Khoury, L. (2021). A poorly understood disease? The impact of COVID-19 on the income gradient in mortality over the course of the pandemic. *European Economic Review*, 140, 103923. https://doi.org/10.1016/j.euroe corey 2021 103923
- Cameron, L., Suarez, D. C., & Rowell, W. (2019). Female labour force participation in Indonesia: Why has it stalled? Bulletin of Indonesian Economic Studies, 55 (2), 157–192. https://doi.org/10.1080/00074918. 2018.1530727
- Chung, Y., & Maguire-Jack, K. (2020). Understanding movement into poverty and poverty persistence over time. Journal of Poverty, 24(3), 241–255. https://doi. org/10.1080/10875549.2019.1692271
- Dang, H.-A. H., & Dabalen, A. L. (2019). Is poverty in Africa mostly chronic or transient? Evidence from synthetic panel data. *The Journal of Development Studies*, 55 (7), 1527–1547. https://doi.org/10.1080/00220388. 2017.1417585

Dang, H.-A. H., & Lanjouw, P. F. (2020). Welfare dynamics in India over a quarter century.

- Dartanto, T., Moeis, F. R., & Otsubo, S. (2020). Intragenerational Economic mobility in Indonesia: A transition from poverty to the middle class in 1993–2014. Bulletin of Indonesian Economic Studies, 56(2), 193–224. https://doi.org/10.1080/00074918. 2019.1657795
- Dartanto, T., & Nurkholis. (2013). The determinants of poverty dynamics in Indonesia: Evidence from panel data. Bulletin of Indonesian Economic Studies, 49(1), 61–84. https://doi.org/10.1080/00074918.2013. 772939
- Dartanto, T., Yuan, E. Z. W., & Sofiyandi, Y. (2018). Structural transformation and the dynamics of income equality in Indonesia: 1996-2014. Kuznets beyond Kuznets, structural transformation and income inequality in the Era of globalization in Asia. In S. Paul (Ed.), Asian Development Bank Institute. https://www.adb.org/sites/default/files/publication/ 451936/adbi-kuznets-beyond-kuznets-incomeinequality-globalization-asia.pdf#page=185
- De Silva, I. (2008). Micro-level determinants of poverty reduction in Sri Lanka: A multivariate approach. International Journal of Social Economics, 35(3), 140–158. https://doi.org/10.1108/ 03068290810847833
- De Silva, I., & Sumarto, S. (2015). Dynamics of growth, poverty and human capital: Evidence from Indonesian Sub-National data. *Journal of Economic Development*, 40(2), 1. https://doi.org/10.35866/ caujed.2015.40.2.001
- Deyshappriya, N. R., & Minuwanthi, R. (2020). Determinants of poverty: Is age non-linearly related with poverty? Evidence from Sri Lanka. International Journal of Asian Social Science, 10(4), 181–192. https://doi.org/10.18488/journal.1.2020.104.181.192
- Duclos, J.-Y., Araar, A., & Giles, J. (2010). Chronic and transient poverty: Measurement and estimation, with evidence from China. *Journal of Development Economics*, 91(2), 266–277. https://doi.org/10.1016/j. jdeveco.2009.09.002
- Esquivias, M. A., Jayanti, A. D., Sethi, N., & Ramandha, M. D. (2020). Financial inclusion dynamics in Southeast Asia: An empirical investigation on three countries. *Business Strategy & Development*, 4(2), 203–215. https://doi.org/10.1002/ bsd2.139
- Esquivias, M. A., Sugiharti, L., Jayanti, A. D., Purwono, R., & Sethi, N. (2020). Mobile technologies, financial inclusion and inclusive growth in East Indonesia. *Journal* of Telecommunications and the Digital Economy, 8(2), 123–145. https://doi.org/10.18080/jtde.v8n2.253
- Farooq, S., & Ahmad, U. (2020). Economic growth and rural poverty in Pakistan: A panel dataset analysis. *European Journal of Development Research*, 32(4), 1128–1150. https://doi.org/10.1057/s41287-020-00259-y
- Fernández-Ramos, J., Garcia-Guerra, A. K., Garza-Rodriguez, J., & Morales-Ramirez, G. (2016). The dynamics of poverty transitions in Mexico. International Journal of Social Economics, 43(11), 1082–1095. https://doi.org/10.1108/IJSE-04-2015-0084
- Firdausy, C. M., & Budisetyowati, D. A. (2022). Variables, dimensions, and Indicators important to develop the multidimensional poverty line measurement in Indonesia. Social Indicators Research, 162(2), 763–802. https://doi.org/10.1007/s11205-021-02859-5

- Fitrinitia, I. S., & Matsuyuki, M. (2022). Role of social protection in coping strategies for floods in poor households: A case study on the impact of program Keluarga Harapan on labor households in Indonesia. International Journal of Disaster Risk Reduction, 80, 103239. https://doi.org/10.1016/j.ijdrr.2022.103239
- García-Vélez, D. F., Quezada-Ruiz, L. D., Tituaña-Castillo, M. D. C., & Del Río-Rama, M. D. L. C. (2022). Regional analysis of poverty in Ecuador: Sensitivity to the choice of equivalence scales. Frontiers in Psychology, 13, 877427. https://doi.org/ 10.3389/fpsyg.2022.877427
- Garza-Rodriguez, J., Ayala-Diaz, G. A., Coronado-Saucedo, G. G., Garza-Garza, E. G., & Ovando-Martinez, O. (2021). Determinants of poverty in Mexico: A quantile regression analysis. *Economies*, 9(2), 60. https://doi. org/10.3390/economies9020060
- Garza-Rodriguez, J., Gonzalez-Martinez, M., Quiroga-Lozano, M., Solis-Santoyo, L., & Yarto-Weber, G. (2010). Chronic and transient poverty in Mexico: 2002-2005. Economics Bulletin, 30(4), 3188–3200.
- Gibson, J., Jiang, Y., & Susantono, B. (2023). Revisiting the role of secondary towns: How different types of urban growth relate to poverty in Indonesia. World Development, 169, 106281. https://doi.org/10.1016/j. worlddev.2023.106281
- Glewwe, P. (2012). How much of observed economic mobility is measurement error? IV methods to reduce measurement error bias, with an application to Vietnam. The World Bank Economic Review, 26(2), 236–264. https://doi.org/10.1093/wber/lhr040
- Glewwe, P., & Twum-Baah, A. (1991). The Distribution of Welfare in Ghana, 1987–1988. World Bank Living Standards Measurement Study Working Paper, 75. The World Bank: Washington, DC, USA.
- Hanandita, W., & Tampubolon, G. (2016). Multidimensional poverty in Indonesia: Trend over the last decade (2003–2013). *Social Indicators Research*, 128(2), 559–587. https://doi.org/10.1007/ s11205-015-1044-0
- Hauser, I. (2023). Welfare, state, and values: The winding road of the normative approach to inequality measurement (1912–1970). The European Journal of the History of Economic Thought, 30(5), 832–859. https:// doi.org/10.1080/09672567.2023.2248309
- Ichwara, J. M., Kiriti-Ng'ang'a, T. W., & Wambugu, A. (2023). Changes in gender differences in household poverty in Kenya. Cogent Economics & Finance, 11(1), 2191455. https://doi.org/10.1080/23322039.2023. 2191455
- Jalan, J., & Ravallion, M. (1998). Transient poverty in Postreform rural China. *Journal of Comparative Economics*, 26(2), 338–357. https://doi.org/10.1006/ jcec.1998.1526
- Jalan, J., & Ravallion, M. (2000). Is transient poverty different? Evidence for rural China. *The Journal of Development Studies*, 36(6), 82–99. https://doi.org/10. 1080/00220380008422655
- Janz, T., Augsburg, B., Gassmann, F., & Nimeh, Z. (2023). Leaving no one behind: Urban poverty traps in sub-Saharan Africa. *World Development*, 172, 106388. https://doi.org/10.1016/j.worlddev.2023. 106388
- Khosla, S., Jena, P. R., & Rahut, D. B. (2023). Can rural livelihood programs enhance capabilities and reduce vulnerability to poverty? Evidence from a tribal region of eastern India. *Economic Analysis and Policy*, 77, 85–98. https://doi.org/10.1016/j.eap.2022.10.018
- Kis-Katos, K., & Sparrow, R. (2015). Poverty, labor markets and trade liberalization in Indonesia. *Journal of*

Development Economics, 117, 94–106. https://doi. org/10.1016/j.jdeveco.2015.07.005

- Koenker, R., & Bassett, G. (1978). Regression quantiles. *Econometrica*, 46(1), 33. https://doi.org/10.2307/ 1913643
- Krishna, A., & Shariff, A. (2011). The irrelevance of National strategies? Rural poverty dynamics in States and regions of India, 1993–2005. World Development, 39(4), 533–549. https://doi.org/10. 1016/j.worlddev.2010.08.011
- Laajaj, R., Webb, D., Aristizabal, D., Behrentz, E., Bernal, R., Buitrago, G., Cucunubá, Z., De La Hoz, F., Gaviria, A., Hernández, L. J., De Los Rios, C., Ramírez Varela, A., Restrepo, S., Schady, N., & Vives, M. (2022). Understanding how socioeconomic inequalities drive inequalities in COVID-19 infections. *Scientific Reports*, 12(1), 8269. https://doi.org/10.1038/s41598-022-11706-7
- Lagakos, D. (2020). Urban-rural gaps in the developing world: Does internal migration offer opportunities? *Journal of Economic Perspectives*, 34(3), 174–192. https://doi.org/10.1257/jep.34.3.174
- Landiyanto, E. A. (2021). Comparison of child poverty measures: Looking for consensus. *Child Indicators Research*, 15(1), 35–66. https://doi.org/10.1007/ s12187-021-09867-4
- Leal Filho, W., Henrique Paulino Pires Eustachio, J., Dinis, M. A. P., Sharifi, A., Venkatesan, M., Donkor, F. K., Doni, F., Abubakar, I. R., Cichos, K., & Vargas-Hernández, J. (2022). Transient poverty in a sustainable development context. International Journal of Sustainable Development & World Ecology, 29(5), 415–428. https://doi.org/10.1080/13504509. 2022.2029612
- Lee, N., Ridder, G., & Strauss, J. (2017). Estimation of poverty transition matrices with noisy data. *Journal* of Applied Econometrics, 32(1), 37–55. https://doi.org/ 10.1002/jae.2506
- Leeuwen, B. V., & Földvári, P. (2016). The development of inequality and poverty in Indonesia, 1932–2008. Bulletin of Indonesian Economic Studies, 52(3), 379–402. https:// doi.org/10.1080/00074918.2016.1184226
- Lo Bue, M. C., & Palmisano, F. (2020). The individual poverty incidence of growth. Oxford Bulletin of Economics and Statistics, 82(6), 1295–1321. https:// doi.org/10.1111/obes.12362
- Mahadevan, R., Nugroho, A., & Amir, H. (2017). Do inward looking trade policies affect poverty and income inequality? Evidence from Indonesia's recent wave of rising protectionism. *Economic Modelling*, *62*, 23–34. https://doi.org/10.1016/j.econmod.2016.12.031
- Mai, T., & Mahadevan, R. (2016). A research note on the poverty dynamics and cost of poverty inequality: Case study of Indonesia. *Economic Analysis and Policy*, 49, 100–107. https://doi.org/10.1016/j.eap. 2015.12.003
- Mehari, A. T. (2022). Prevalence of poverty and inequality in Ethiopia: Using Ethiopian Socio-Economic Survey (ESS) 2013 and 2015. Journal of Poverty, 26(5), 385–402. https://doi.org/10.1080/10875549.2021. 1925802
- Mendola, D., & Busetta, A. (2012). The Importance of consecutive spells of poverty: A Path-dependent index of longitudinal poverty. *Review of Income and Wealth*, 58(2), 355–374. https://doi.org/10.1111/j. 1475-4991.2011.00486.x
- Moeis, F. R., Dartanto, T., Moeis, J. P., & Ikhsan, M. (2020). A longitudinal study of agriculture households in Indonesia: The effect of land and labor mobility on welfare and poverty dynamics. *World Development*

Perspectives, 20, 100261. https://doi.org/10.1016/j. wdp.2020.100261

- Mulyoutami, E., Lusiana, B., & van Noordwijk, M. (2020). Gendered migration and agroforestry in Indonesia: Livelihoods, labor, know-how, networks. *Land*, *9*(12), 529. https://doi.org/10.3390/land9120529
- Muryani, N. A., & Esquivias, M. A. (2021). Factors influencing the gender gap in poverty: The Indonesian case. World Review of Entrepreneurship Management and Sustainable Development, 17(1), 103–119. https://doi. org/10.1504/WREMSD.2021.112101
- Muyanga, M., Jayne, T. S., & Burke, W. J. (2013). Pathways into and out of poverty: A study of rural household wealth dynamics in Kenya. *The Journal of Development Studies*, 49(10), 1358–1374. https://doi. org/10.1080/00220388.2013.812197
- Nawab, T., Raza, S., Shabbir, M. S., Yahya Khan, G., & Bashir, S. (2023). Multidimensional poverty index across districts in Punjab, Pakistan: Estimation and rationale to consolidate with SDGs. *Environment Development and Sustainability*, 25(2), 1301–1325. https://doi.org/10.1007/s10668-021-02095-4
- Noerhidajati, S., Purwoko, A. B., Werdaningtyas, H., Kamil, A. I., & Dartanto, T. (2020). Household financial vulnerability in Indonesia: Measurement and determinants. *Economic Modelling*, 96, S433-444. https://doi.org/10.1016/j.econmod.2020.03.028
- Özsoy, Ö., & Gürler, M. (2022). Poverty and social exclusion of older people in ageing European Union and Turkey. Journal of Public Health, 30(8), 1969–1983. https://doi.org/10.1007/s10389-021-01525-x
- Peng, C., Fang, L., Wang, J. S.-H., Law, Y. W., Zhang, Y., & Yip, P. S. F. (2019). Determinants of poverty and their variation across the poverty spectrum: Evidence from Hong Kong, a high-income society with a high poverty level. Social Indicators Research, 144(1), 219–250. https://doi.org/10.1007/s11205-018-2038-5
- Poy, S. (2023). In-work poverty dynamics: Trigger events and short-term trajectories in Argentina. *Journal for Labour Market Research*, 57(1), 24. https://doi.org/10. 1186/s12651-023-00348-5
- Purwono, R., Wardana, W. W., Haryanto, T., & Khoerul Mubin, M. (2021). Poverty dynamics in Indonesia: Empirical evidence from three main approaches. World Development Perspectives, 23, 100346. https:// doi.org/10.1016/j.wdp.2021.100346
- Putri, S. R., Wijayanto, A. W., & Sakti, A. D. (2022). Developing relative spatial poverty index using integrated remote sensing and geospatial big data approach: A case study of East Java, Indonesia. *ISPRS International Journal of Geo-Information*, 11(5), 275. https://doi.org/10.3390/ijgi11050275
- Radeny, M., Van Den Berg, M., & Schipper, R. (2012). Rural poverty dynamics in Kenya: Structural declines and stochastic escapes. World Development, 40(8), 1577–1593. https://doi.org/10.1016/j.worlddev.2012. 04.027
- Roy, P., Ray, S., & Haldar, S. K. (2019). Socio-economic determinants of multidimensional poverty in rural West Bengal: A household level analysis. *Journal of Quantitative Economics*, 17(3), 603–622. https://doi. org/10.1007/s40953-018-0137-4
- Ruggeri Laderchi, C., Spatafora, N., Shetty, S., & Zaidi, S. (2017). Riding the wave: An East Asian Miracle for the 21st Century. The World Bank. https://doi.org/10. 1596/978-1-4648-1145-6
- Sahara, D. A., Amaliah, S., Irawan, T., & Dilla, S. (2022). Economic impacts of biodiesel policy in Indonesia: A computable general equilibrium approach. *Journal*

of Economic Structures, 11(1), 22. https://doi.org/10. 1186/s40008-022-00281-9

- Schaner, S., & Das, S. (2016). Female labor force participation in Asia: Indonesia country study. Asian Development Bank Economics Working Paper Series, 474. https://doi.org/10.2139/ssrn.2737842
- Schotte, S., Zizzamia, R., & Leibbrandt, M. (2022). Snakes and ladders and loaded dice: Poverty dynamics and inequality in South Africa between 2008 and 2017. South African Journal of Economics, 90(2), 214–242. https://doi.org/10.1111/saje.12308
- Shabnam, N., Ameer, W., Aurangzeb, N., Ashraf, M. A., Shah, S. H., & Marchetti, S. (2023). Estimation of poverty bounds for Pakistan using synthetic panel data. *PLOS ONE*, 18(3), e0276673. https://doi.org/10. 1371/journal.pone.0276673
- Shah, S., & Debnath, N. (2022). Determinants of multidimensional poverty in rural Tripura, India. Journal of Quantitative Economics, 20(1), 69–95. https://doi.org/ 10.1007/s40953-021-00256-w
- Sparrow, R., Dartanto, T., & Hartwig, R. (2020). Indonesia under the new normal: Challenges and the way ahead. Bulletin of Indonesian Economic Studies, 56 (3), 269–299. https://doi.org/10.1080/00074918. 2020.1854079
- Sparrow, R., Suryahadi, A., & Widyanti, W. (2013). Social health insurance for the poor: Targeting and impact of Indonesia's askeskin programme. Social Science & Medicine, 96, 264–271. https://doi.org/10.1016/j.socs cimed.2012.09.043
- Sugiharti, L., Aditina, N., & Esquivias, M. A. (2022). Worker transition across formal and informal sectors: A Panel data analysis in Indonesia. Asian Economic and Financial Review, 12(11), 923–937. https://doi. org/10.55493/5002.v12i11.4635
- Sugiharti, L., Purwono, R., Esquivias, M. A., & Jayanti, A. D. (2022). Poverty dynamics in Indonesia: The prevalence and causes of chronic poverty. *Journal of Population and Social Studies*, 30, 423–447. https:// doi.org/10.25133/JPSSv302022.025
- Sugiharti, L., Purwono, R., Esquivias, M. A., & Rohmawati, H. (2023). The nexus between crime rates, poverty, and income inequality: A case study of Indonesia. *Economies*, 11(2), 62 https://doi.org/10.3390/ economies11020062
- Sundar Pani, B., & Mishra, D. (2022). Sustainable livelihood security in Odisha, India: A district level analysis.

Regional Sustainability, 3(2), 110–121. https://doi.org/ 10.1016/j.regsus.2022.07.003

- Taufiq, N., & Dartanto, T. (2020). Education, informal turnover and poverty dynamics in Indonesia. International Journal of Economics & Management, 14(1), 157–172.
- Tohari, A., Parsons, C., & Rammohan, A. (2019). Targeting poverty under complementarities: Evidence from Indonesia's unified targeting system. *Journal of Development Economics*, 140, 127–144. https://doi. org/10.1016/j.jdeveco.2019.06.002
- Tsiboe, F., Armah, R., Zereyesus, Y. A., & Annim, S. K. (2023). Food poverty assessment in Ghana: A closer look at the spatial and temporal dimensions of poverty. *Scientific African*, 19, e01518. https://doi.org/ 10.1016/j.sciaf.2022.e01518
- Usman, H., & Projo, N. W. K. (2021). Encouraging Entrepreneurship for people with disabilities in Indonesia: The United Nations"Leave No one Behind'Promise. Journal of Population and Social Studies [JPSS], 29, 195–206. https://doi.org/10.25133/ JPSSv292021.012
- Vidyattama, Y. (2013). Regional convergence and the role of the neighbourhood effect in decentralised Indonesia. Bulletin of Indonesian Economic Studies, 49(2), 193–211. https://doi.org/10.1080/00074918. 2013.809841
- Villa, J. M., & Niño-Zarazúa, M. (2019). Poverty dynamics and graduation from conditional cash transfers: A transition model for Mexico's progresaoportunidades-prospera program. *Journal of Economic Inequality*, 17(2), 219–251. https://doi.org/ 10.1007/s10888-018-9399-5
- Widyanti, W., Suryahadi, A., Sumarto, S., & Yumna, A. (2009). The Relationship Between Chronic Poverty and Household Dynamics: Evidence from Indonesia. Poverty Research Centre Working Paper. No. 132. https://dx.doi.org/10.2139/ssrn.1537080
- Yassine, A., & Bakass, F. (2022). Do education and employment play a role in youth's poverty alleviation? Evidence from Morocco. Sustainability, 14(18), 11750. https://doi.org/10.3390/su141811750
- Zhang, W., Xu, W., & Wang, X. (2019). Regional convergence clubs in China: Identification and conditioning factors. The Annals of Regional Science, 62(2), 327–350. https://doi.org/10.1007/s00168-019-00898-y